Author’s response to reviews

Title: Decision-making about antidepressant medication use in pregnancy: a comparison between women making the decision in the preconception period versus in pregnancy

Authors:

Lucy Barker (lucy.barker@utoronto.ca)
Cindy-Lee Dennis (cindylee.dennis@utoronto.ca)
Neesha Hussain-Shamsy (neesha.hussain.shamsy@gmail.com)
Donna Stewart (donna.stewart@uhn.ca)
Sophie Grigoriadis (Sophie.Grigoriadis@sunnybrook.ca)
Kelly Metcalfe (kelly.metcalfe@utoronto.ca)
Timothy Oberlander (toberlander@bcchr.ca)
Carrie Schram (Carrie.Schram@wchospital.ca)
Valerie Taylor (valerie.taylor3@albertahealthservices.ca)
Simone Vigod (simone.vigod@wchospital.ca)

Version: 2 Date: 17 Jan 2020

Author’s response to reviews:

Dr. Byrne
Editor-in-Chief
BMC Psychiatry

January 17, 2020
Dear Dr. Byrne,

Thank you for so rapidly reviewing our revised manuscript: “Decision-making about antidepressant medication use in pregnancy: a comparison between women making the decision in the preconception period versus in pregnancy”. We greatly appreciate the valuable feedback and the opportunity to further revise our work.

Below, please find our response to reviewer comments. We have included the original reviewer comments in {} and our response follows the -&gt; sign. We have included changes to the manuscript within our response. A track changes manuscript is attached.

We hope that these changes meet your and the reviewers’ expectations, and that the changes will render the manuscript suitable for publication in BMC Psychiatry.

Sincerely, and on behalf of all co-authors,

Lucy Barker, MD, FRCPC